Author's response to reviews

**Title:** Inflammatory fibroid polyp of the ileum presenting with small bowel obstruction in an adult patient: a case report

**Authors:**

Toygar Toydemir (toygartoydemir@hotmail.com)

**Version:** 4 **Date:** 30 May 2010

**Author's response to reviews:**

1-The ethnicity of the patient was added to the abstract section.
2-The title of the paper was removed from the abstract page.
3-The style of the title words was changed.